Author's response to reviews

Title: Characteristics of control group participants who increased their physical activity in a cluster-randomized lifestyle intervention trial

Authors:

Lauren A Waters (l.waters1@uq.edu.au)
Marina M Reeves (m.reeves@sph.uq.edu.au)
Brianna S Fjeldsoe (b.fjeldsoe@sph.uq.edu.au)
Elizabeth G Eakin (e.eakin@sph.uq.edu.au)

Version: 3 Date: 14 December 2010

Author's response to reviews:

Thank you for the opportunity to respond to reviewer comments on our manuscript (1350393895395144), entitled “Characteristics of control group participants who increased their physical activity in a cluster-randomized lifestyle intervention trial.” Below, we detail our response to each comment, with corresponding changes highlighted in the manuscript.

We believe that the changes made in response to the reviewers’ comments have improved the manuscript, and hope that the editorial team will now find it suitable for publication in BMC Public Health.